Author's response to reviews

Title: Preventing childhood obesity during infancy in UK primary care: a mixed-methods study of HCPs' knowledge, beliefs and practice

Authors:

Sarah A Redsell (Sarah.Redsell@nottingham.ac.uk)
Philippa J Atkinson (p.j.atkinson@bopenworld.com)
Dilip Nathan (dilip.nathan@nhs.net)
Aloysius N Siriwardena (nsiriwardena@lincoln.ac.uk)
Judy A Swift (judy.swift@nottingham.ac.uk)
Cris Glazebrook (cris.glazebrook@nottingham.ac.uk)

Version: 8 Date: 22 June 2011

Author's response to reviews:

I have added the words health care professionals (HCPs) in full the first time it appears in the manuscript (it was already in full the first time it appeared in the Abstract). I have also put spaces in between the text and the numbered references as this was not consistent. I have also slightly changed the wording in the first theme of the qualitative text from:

Participants considered that some infants were over fed, particularly those on formula milk. They gave detailed accounts of how parents ended up over feeding their infants.

to: Participants did not identify infants who were obese but considered that some were over fed, particularly those on formula milk who were weaned early. The new version fits the quote much better.

Thank you for publishing this paper.